### Lawrence Berkeley National Laboratory

LBL Publications

#### Title

Vulnerability and resilience of urban energy ecosystems to extreme climate events: A systematic review and perspectives

Permalink https://escholarship.org/uc/item/79c2r0k3

Authors Perera, ATD Hong, Tianzhen

Publication Date 2023-03-01

DOI 10.1016/j.rser.2022.113038

Copyright Information

This work is made available under the terms of a Creative Commons Attribution-NonCommercial License, available at <u>https://creativecommons.org/licenses/by-nc/4.0/</u>

Peer reviewed



Building Technologies & Urban Systems Division Energy Technologies Area Lawrence Berkeley National Laboratory

### Vulnerability and resilience of urban energy ecosystems to extreme climate events: A systematic review and perspectives

A. T. D. Perera and Tianzhen Hong

Building Technology and Urban Systems Division, Lawrence Berkeley National Laboratory

Energy Technologies Area December 2022

DOI: 10.1016/j.rser.2022.113038



This work was supported by the Assistant Secretary for Energy Efficiency and Renewable Energy, Building Technologies Office, of the US Department of Energy under Contract No. DE-AC02-05CH11231.

#### Disclaimer:

This document was prepared as an account of work sponsored by the United States Government. While this document is believed to contain correct information, neither the United States Government nor any agency thereof, nor the Regents of the University of California, nor any of their employees, makes any warranty, express or implied, or assumes any legal responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by its trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof, or the Regents of the University of California. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof or the Regents of the University of California.

# Vulnerability and resilience of urban energy ecosystems to extreme climate events: A systematic review and perspectives

#### A. T. D. Perera<sup>1</sup> and Tianzhen Hong

Building Technology and Urban Systems Division Lawrence Berkeley National Laboratory 1 Cyclotron Road, Berkeley, CA, 94720, United States

#### Abstract

We reviewed the present studies on the vulnerability and resilience of the energy ecosystem (most parts of the energy ecosystem), considering extreme climate events. This study revealed that the increased interactions formed during the transformation of the energy landscape into an ecosystem could notably increase the vulnerability of the energy infrastructure. Such complex ecosystem cannot be assessed using the present state of the art models used by the energy system modelers. Therefore, this study introduces a novel analogy known as the COVID analogy to understand the propagation of disruption within and beyond the energy ecosystem and organized the present state of the art based on the COVID analogy. The analogy helps to categorize the vulnerability of the energy infrastructure into three stages. The study revealed that although there are many publications covering the vulnerability and resilience of the energy infrastructure, considering extreme climate events, the majority are focused on the direct impact of extreme climate on the energy ecosystem. In addition, most of the studies do not consider the impact of future climate variations during this assessment. The propagation of disruptions was assessed mainly for wildfires and hurricanes. Further, there is a clear research gap in considering vulnerability assessment for interconnected energy infrastructure. The transformation of energy systems into a complex ecosystem notably increases the complexity, making it difficult to assess vulnerability and resilience. A shift from a centralized to decentralized modeling architecture could be beneficial when considering the complexities brought by that transformation. Hybrid models consisting of both physical and data-driven machine learning techniques could also be beneficial in this context.

#### Highlights

- An exponential increase in publications covering the vulnerability and resilience of the energy infrastructure, especially for extreme climate events
- A novel analogy known as the *COVID analogy is introduced to* classify the state of the art while understanding the propagation of disruption within and beyond the energy ecosystem
- Transformation of the energy landscape into an ecosystem could notably increase the vulnerability during extreme events
- There is a clear research gap in considering vulnerability assessment for interconnected energy infrastructure

**Keywords**: Urban system, energy ecosystem, climate change, extreme climate event, resilience, modeling

<sup>&</sup>lt;sup>1</sup> Corresponding Author. Email: atdasun@gmail.com, atperera@lbl.gov

#### 1. Introduction

Cities are growing at a rapid speed, and are centers of energy consumption and carbon dioxide  $(CO_2)$  emissions [1]. More and more people tend to move into urban areas, increasing urban densities (vertical and horizontal growth). It is expected that global urban coverage will increase by 40% in 2050, adding several billions of people [2]. A significant change in activities linked to commercial and industrial processes takes place within the cities, which are hubs to social and economic growth. As a result of such activities,  $CO_2$  impacts will notably increase; it is predicted that cities will contribute to 70% of the global  $CO_2$  emissions in 2050 [3]. Therefore, improving the sustainability of cities plays a major role in reducing  $CO_2$  emissions.

The energy sector is the main contributor to  $CO_2$  emissions in urban areas, therefore, that sector requires a major transition [1]. Large-scale integration of renewable energy technologies occurs at the building level and direct grid integration of renewables improves sustainability [4,5]. Solar and wind energy plays a significant role in substituting fossil fuel-based energy technologies in the electricity sector. These technologies are non-dispatchable, as their energy generation is strongly influenced by the weather conditions, in contrast to fossil fuel-based systems where controllers can control the power generation much more effectively [6]. As a result, auxiliary support through energy storage and demand response strategies are needed to improve the penetration level of these technologies [6]. In addition to transforming the electricity sector, renewable energy technologies must play a major role in transforming the building and transportation sectors [7]. Transforming the fossil fuel-based heating sector to an electricity-based sector while accommodating increasing cooling demand, especially due to climate change and urban heat islands, makes the challenge even greater [8,9]. Similarly, an increasing number of electric vehicles depend on the grid. It is expected that major part of the transportation sector will consist of electric vehicles in the future [10].

Inclusion of the building and transportation sectors in overall energy infrastructure planning, design, and operation has certain advantages [9,11]. These sectors can substantially enhance the flexibility of the energy infrastructure, which can help to increase the penetration level of non-dispatchable renewable energy technologies. Furthermore, coupling building and transportation sectors with the energy infrastructure significantly enhances the electricity market, creating more opportunities (job creation and technology development) in the energy sector and speeding the energy transition [12]. However, catering to the energy demand for building and transportation sectors while going through the energy transition is difficult. Coupling energy infrastructure with the building and transportation sectors will create a complex ecosystem, especially when using the flexibility of building and transportation in one sector could quickly propagate and lead to a blackout in the entire ecosystem [13,14]. Therefore, the reliability and resilience of the urban energy ecosystem need to be enhanced while going through the transition.

Extreme climate events driven by climate change have a notable impact on the transformation of the energy ecosystem [15]. The frequency and intensity of extreme climate events has notably increased during the last decade and will continue to increase further [1]. These events are having a multidimensional impact on the energy ecosystem [13]. For example, the recent extreme cold event in Texas directly affected the building stock, increasing heating energy demand significantly [16]. At the same time, the extremely low temperatures interrupted the operation of gas turbines and wind turbines. Mismatch in demand and generation at the different parts of the state collapsed the transmission and distribution, leading to a rolling blackout, causing a loss of 130 billion dollars [16] and resulting in more than 200 deaths [17]. The compound impact of extreme climate events on the entire energy ecosystem makes it quite challenging to improve resilience [14,15]. Therefore, it is important to take special measures to improve resilience while improving interconnectivity.

Therefore, climate change mitigation and adaptation make it essential to improve sustainability, interconnectivity, and resilience, which is a major challenge human civilization has to face.

There has been an ongoing interest in the present state of the art to improve the sustainability, resilience, and interconnectivity in the urban energy ecosystem. A clear trend can be seen since 2015 (Fig. 1). Therefore, it is crucial to review the present state of the art and summarize the work performed and identify the research gaps that can be addressed in the future. Several review articles have also been presented that cover broad topics related to the resilience of cities. A holistic overview of these reviews is presented in Section 6. They are either focused on a top-down approach that addresses the resilience at the urban context or focused on specific areas of the energy ecosystem (further discussed in detail under Section 6). Such holistic assessments do not provide a clear understanding of the cascading impact on an interconnected energy infrastructure. On the other hand, reviews focused on specific areas of the energy ecosystem do not provide an overall impact. Balancing both depth and breadth while organizing the study is a challenging task. Within this context, the authors aim to:

- Introduce a COVID analogy to understand the propagation of disruption in energy ecosystems to help grasp both a holistic and detailed (bottom-up) overview regarding the propagation of disruption due to extreme events.
- Review existing workflows developed to quantify the vulnerability for disruption propagation created by extreme climate events within the energy ecosystem being based on the COVID analogy.
- Review possible descriptions for the resilience, in order to arrive at a more holistic interpretation of resilience considering the energy ecosystem and the propagation of disruption in multiple sectors.
- Highlight promising modeling frameworks in the present state of the art and research gaps to visualize (generate a more holistic viewpoint) resilience and sustainability of the urban energy ecosystem.



Fig. 1: Recent trend in publications considering resilience, sustainability, and interconnectivity in the urban energy infrastructure indexed in the Scopus database.

The review is performed considering the extreme events and urban energy resilience in a more holistic manner. The intersection of extreme events and resilient communities (urban scale) leads to a broad research outlook (Fig. 2) which cannot be reviewed in a single paper. In general, urban resilience is a result of resilient networks and flows, resilient urban infrastructure and form, resilient governance and network, and resilient socioeconomics and dynamics (Fig. 2). Within this paper we limit our scope to networked materials and energy flows (energy flow is given a major priority) and urban infrastructure and form. The broad approach we used within the urban context enables us to move beyond the boundaries of energy systems and consider the urban energy ecosystem. Within this context, we used several combinations of the following keywords and performed a literature review in Scopus. The keywords include climate change, extreme events, urban resilience, network cascade failures, infrastructure, energy systems, vulnerability, wildfires, hurricanes, urban resilience, energy resilience, building energy, extreme heat events, and interconnectivity.



## Fig. 2: scope of the study within the broad umbrella of urban resilience and extreme events focusing on urban energy ecosystem

The paper is organized as follows: Section 2 presents a compressive overview of the changes taking place in the energy sector. Section 3 presents a novel analogy to understand the vulnerability of energy infrastructure for extreme events. Section 4 presents the direct impact of extreme events on the energy infrastructure. Section 5 presents the propagation of such extreme events within and beyond the energy infrastructure. Section 6 presents the limitations within the present state of the art for understanding the impacts brought by these extreme events. Section 7 presents areas of interest for future research activities to improve the resilience of energy ecosystems. Finally, conclusions are presented in Section 8.

#### 2. Transition in the urban energy sector

The energy sector is going through a major transition, where renewable energy technologies replace traditional power generation technologies based on fossil fuels. Power generation using fossil fuels based on the Rankine cycle, the Bryton Cycle, or internal combustion generators are mature technologies that have been amply discussed in the literature [18]. Similarly, the literature has amply discussed the concurrent generation of heat and electricity using fossil fuel resources with a vapor power cycle and a gas power cycle. However, this should not undermine recent efforts to develop thermodynamic cycles such as the Organic Rankine Cycle, the Kalina cycle, and others. These are live examples of improvements in energy conversion methods [19,20].

Recent trends to reduce the carbon impact has led to the introduction of geothermal, biomass, and bioenergy to replace fossil fuel resources. However, a notable change in energy system architecture cannot be observed in these instances, although improvements in energy conversion technologies are reported (except for hybrid power cycles). Similar to renewable-based thermal energy, grid integrated solar photovoltaic (PV) systems and wind farms have been amply discussed, especially due to the pressure brought by climate change and the uncertainty in fossil fuel availability [21]. Integration of PV and wind turbines has increased the complexity of the energy infrastructure due to their intermittent nature [22]. Hybrid energy systems consisting of battery banks and dispatchable energy technologies are often used to withstand these fluctuations locally [22,23]. This leads to the emergence of distributed energy systems, and popularized by concepts such as energy hubs [24]. Requirements to decarbonize the building, industry, and transportation sectors add burden and further increase complexity. Several system configurations have been proposed to provide multi-energy services [12], depending on the sectors coupled with the energy infrastructure. Energy markets facilitate the interaction between multiple sectors while maintaining a reliable supply. Participation in many sectors makes a complex landscape that leads to an ecosystem, as shown in Fig. 3.

These connections between different participants through multiplex networks creates a complex ecosystem which can be extremely vulnerable to extreme events. Therefore, maintaining a robust and resilient service in such an ecosystem is a challenging task. As a result, we may need to move beyond typical energy system models to capture these interactions. The ecosystem concept becomes quite ideal in such a context which has already been formulated in the context of industrial ecosystem, business ecosystem, digital ecosystem, and innovation ecosystem [25]. The interactions between different actors, taking into account energy and material flows within the urban context, makes industrial ecosystem models do not consider detailed bottom-up models used in the energy system domain. Within this context, energy ecosystems rely on co-existence of different actors within the urban context including building, industrial, commercial and transportation sectors. Each sector may try to utilize the materials, energy (heat, electricity, gas) in a manner that the waste, dependence beyond the boundaries, and environmental impact will be minimal.



Fig. 3: Formulation of a complex energy ecosystem: (a) the transformation in the energy infrastructure starting from dispatchable energy sources, dispatchable energy sources with non-dispatchable energy technologies, dispatchable energy sources with non-dispatchable energy technologies with energy storage, and finally becoming an ecosystem being connected to buildings (commercial, residential, and industrial) and electric vehicles with demand response strategies. (b) the interconnections within an energy ecosystem. (c) further elaboration of the interconnections between the energy ecosystem through multiplex networks including energy and cyber interactions. The interconnection between different layers highlights the interdependency between different layers. For example, the functioning of electricity depends on the support by the communication layer in order to maintain a stable electricity grid. On the other hand, communication layer depends on the electricity grid to obtain energy to function.

# **3.** Moving beyond the domino effect: A COVID analogy to understand the vulnerability of energy ecosystems

The domino effect/theory has been widely used to demonstrate the catastrophic chain of events triggered by a simple event. The domino effect brought by climate change has been amply discussed in the present state of the art. Understanding the propagation of a chain of events becomes more challenging when it comes to the energy domain, due to complex interactions between different energy carriers, as well as the interactions with the other service sectors at the urban scale. Therefore, defining the resilience, as well as understanding the vulnerabilities, become challenging. Discussions in Section 3.1 and Section 3.2 are devoted to a broader understanding of the concept of resilience and propagation of disruption within the energy ecosystem.

#### 3.1 The concept of resilience

The importance of resilience has been widely discussed recently, mainly due to the disruption in the energy sector, which has been driven by extreme climate events. Nonetheless, the concept of resilience has a much broader scope beyond energy. According to Hamilton [26], resilience is defined as the "ability to recover and continue to provide their main functions of living, commerce, industry, government and social gathering in the face of calamities and other hazards." According to Coaffee [27], resilience has been defined as the "capacity to withstand and rebound from disruptive challenges." Desouza and Flanery [28] introduce resilience as the "ability to absorb, adapt and respond to changes" while Asprone and Latora [29] define resilience as "the capacity to adapt or respond to unusual often radically destructive events." These definitions provide a much more holistic understanding of resilience.

A reasonable change in the definition can be seen when moving from the general expression of resilience into the resilience of energy infrastructure. More importantly, the concept of resilience has been associated quite closely with reliability, flexibility, stability, and robustness [14]. According to Perera et al. [30] *flexibility* has been defined as the capability of a system to withstand the disturbances brought by the external forces with a minimum impact, while *reliability* [31] is defined as the capability to cater the services with a minimum disruption due to the changes in both internal/external conditions. Nik et al. introduced *resilience* as the capability of energy infrastructure to be flexible and reliable during high-probable low-impact events [14]. Besides being flexible and reliable, being able to restore a system to stable operation following a disruption is also an important characteristic of a resilient system [32]. And last, but not least, resilience of energy infrastructure demonstrates the capability to be predictable when responding to extreme events (partly known as *robustness*).

Given the breadth of these definitions, it is clear that the resilience of energy infrastructure is a broad concept closely linked to a group of other concepts. More importantly, the definition is highly specific to the application considered, as well as to the nature of extreme events. Extreme events could disrupt multiple sectors within the energy ecosystem simultaneously and ultimately collapse the operation of the entire ecosystem. As a result, the entire ecosystem could collapse. Such disruptions can easily penetrate the other interconnected sectors such as buildings, transportation, and industry, leading to a complete blackout, resulting in substantial economic loss. Therefore, vulnerability of the energy ecosystem is closely discussed, along with resilience.

#### 3.2 The analogy of the domino effect to COVID

Although vulnerability and the resilience of the energy infrastructure have been widely discussed, the complexity of energy ecosystems and the diversity of extreme events have made it difficult to arrive at a more holistic model regarding the vulnerability of energy ecosystems for extreme events. The COVID analogy is presented to provide a more holistic understanding of the vulnerability of the energy ecosystem for extreme events (Fig. 4). The overall process is divided into three stages. In Stage 1, the COVID virus infects the respiratory tract, which is the entry point. Similarly, an extreme event could disrupt a particular component or a part of the energy ecosystem (Fig. 5). For example, extreme drought can disturb hydropower generation, and a wildfire season can disturb the distribution/transmission system.

Stage 2 describes the propagation. In the case of the Covid virus, it propagates in the respiratory system and damages the lungs. Similarly, the disruption initiated at a specific sector can easily propagate to the entire ecosystem and disrupt its operation. As a result, the entire electricity sector

can collapse in a particular area. The disruption during Stage 2 is more critical than Stage 1 and may require more time to recover.

Stage 3, where the virus further penetrates and damages the heart, kidneys, and digestive system leading to multiple organ damage beyond the respiratory system, is more destructive. Similarly, the disruption in the energy ecosystem can penetrate beyond the energy infrastructure and start to disrupt other interconnected infrastructure such as transportation, communication, and water supply. Such propagation beyond the boundary of the energy ecosystem is more disruptive and leads to a huge economic loss. Further, it takes a much longer time to recover during such a multi-sector disruption. Therefore, improving resilience to avoid such penetration is vital when designing urban systems. The present state-of-the-art studies on improving the resilience of energy ecosystems for Stage 1 to 3 type of events are reviewed in sections 4 and 5, respectively.



Fig. 4: Analogy of Covid to the human system to extreme events on the urban energy ecosystem.



Fig. 5: Understanding the three stages of the disruptions propagation in an energy ecosystem (inline with the COVID analogy). (a) The complex interactions maintained within the energy ecosystem using a multiplex network (inline with Fig. 3) (blue: energy, orange: gas, purple: heat, green: communication). (b) The layers of the multiplex network maintain interactions with each other when operating. (c) Disruptions taken place at a few locations within the energy layer (Stage 1). (d) Propagation of the disruptions within the energy layer, leading towards a blackout (Stage 2). (e) The blackout penetrating the disruptions into the communication layer due to the dependencies of communication networks on the electricity sector. (f) Subsequently, the disruption propagates in the communication layer. (g) Similarly, it propagates into the other layers, collapsing the operation of the entry ecosystem.

#### 4. Direct impact of extreme events on the energy ecosystem (Stage 1)

Extreme events can have a direct impact on generation, demand, and distribution/transmission. The direct impact on the energy ecosystem may change depending on the characteristics of the extreme event. This section covers three such extreme events—wildfires, hurricanes, and extreme cold/heat events—and reviews the present state-of-the-art studies that quantify their direct impacts on the energy ecosystem (Fig. 6).



Fig. 6: Extreme events considered in the present study.

#### 4.1 Impact of wildfires on energy and built infrastructure

Understanding the impact of wildfires on energy (Table 1) and built infrastructure is a challenging task [33]. There exists a complex coupling between the wildfires, energy, and building infrastructure [34]. First, the emission form construction and energy sectors directly help to speed climate change, leading to increased frequency of extreme hot weather conditions that lead to wildfires. In addition, the electricity grid has been a main root cause in triggering wildfires [35]. Although energy infrastructure is responsible for about 3% of all wildfire ignitions, power lines have ignited four of the twenty largest fires in California and 90% of the fatal bushfires in Australia [35]. Therefore, public safety power shutdowns (PSPS) are often performed during wildfire periods, curtailing the power supply in many areas vulnerable to wildfires [36]. Wildfires usually take place during extremely hot climate conditions when cooling energy demand is notably high. A drop in power supply can easily lead to an increase in the health impacts of heatwaves and may increase the mortality rate [34,37]. During wildfire season, such PSPS events lead to poor air quality in buildings and make it difficult to maintain communication between communities, as the telecommunication towers cannot provide the service without electricity. Therefore, wildfires are having a direct impact on both urban and energy ecosystems.

Quantifying the impact threat imposed by wildfire on the energy infrastructure presents many challenges to the energy community [38,39]. Integrated assessment models that couple climate and ecological models help to capture the wildfire threats in the long run, taking into account future

climate variations. Sathaye et al. [33] present a perspective of linking grid models to climate and ecological models to plan the expansion of the energy grid in a wildfire resilient manner. Nonetheless, bottom-up models linking energy planning with climate and ecological models were not found in the present state of art, mainly due to the complexity that arises when linking climate-wildfire and energy system models (Table 1).

The present state-of-the-art models on energy infrastructure resilience during wildfires mainly focus on modeling and assessing the threat of the electricity grid triggering wildfires and understanding the cost-optimal and reliable strategies for PSPS. In many studies, detailed thermal models are often used to understand the threat of triggering wildfires [36,40,41]. At the same time, Wischkaemper et al. [42] discuss the instrumentational methods to detect the risk of wildfires. Based on the risk assessment, optimal power flow and dispatch problems are formulated to determine the optimal power shut down strategy, as explained in many papers (Table 1). Optimal power flow and dispatch problems are combined to consider the shutdowns in power lines due to the threat of wildfires, known as the optimal power shutdown problem (OPSP) [36,43–45]. OPSP has been formulated as a deterministic operation problem in most instances. Trakas and Hatziargyriou [44] used stochastic models to consider the uncertainties in the OPSP. Renewable energy generation and energy storage are often not considered in the OPSP problem, making it relatively easy for optimization algorithms. For example, only Trakas and Hatziargyriou [44] considered energy storage in the OPSP. Hay and Mohit [46] discuss the impact of wildfires, taking into account the future scenarios for building and transportation energy demand. Furthermore, most of the studies are limited to the operational aspect of the problem. Design improvements that could help to withstand public safety power shutdowns have not been incorporated into the discussion. Pacific Gas and Electricity (PG&E), a utility in California, USA, recently introduced microgrids to support the areas surrounded by wildfires [47,48]. These areas do not need to depend on the transmission network during wildfire periods. Incorporating such microgrids into the existing energy infrastructure would be an interesting area of research. Improving climate resilience in energy infrastructure to cope wildfires is an emerging area of research. However, the frequent occurrence of wildfires will demand much attention on this area of research. It will be essential to move beyond the operational level to discuss the design aspects to improve the resilience of energy infrastructure for wildfires.

#### 4.2 Impact of hurricanes on energy and built infrastructure

Hurricanes are a major climate threat to the U.S. energy sector, and are responsible for 47% of weather-related events. A significant economic loss up to \$95 million with an average of \$12 million per extreme event may occur as a result [49]. These events result in adverse structural damage to the grid, which takes 117.5 hours on average to recover from following an extreme event [49]. More importantly, the frequency of extreme climate events has increased due to climate change. As a result, power disruptions increase by 2% every year [50]. Therefore, it is vital to improve the resilience of energy infrastructure in relation to hurricanes.

A number of recent studies have focused on improving the climate resilience of energy infrastructure to withstand hurricanes. Parker et al. [51] reviewed the present state-of-the-art publications, and Chen et al. [52] presented future perspectives that could shape the research to improve resilience. A comparative assessment of the present state-of-the-art publications that focused on improving climate resilience during hurricanes is presented in Table 2. A majority of the papers on hurricane resilience focused on improving the network resilience of the electricity grid during extreme climate events, which is considered a major challenge, as shown in Table 2. Studies [49,53,54] evaluated the tendency for cascade failures during extreme climate events. Besides being limited to the resilience of the network, the compound impact on both generation and distribution/transmission was

investigated in several studies. For example, the need for optimal power flow and dispatch strategy to be adapted during hurricane events was studied by [53,55].

Most of the present state-of-the-art studies are focused on operating energy infrastructure during hurricanes, limiting the focus to the operational aspects. However, it is essential to design energy infrastructure that can withstand hurricanes with a minimal impact. Bennett et al. [55] considered the energy infrastructure design, taking into account hurricanes where sequential optimization technique is used to derive transition pathways to improve the climate resilience for hurricanes in Puerto Rico. Stochastic models have been used in Dehghani [56]. Addressing the challenge of improving climate resilience at the design stage would be an interesting future direction. Considering n-1 security, incorporating uncertainties introduced by climate, human system, and other factors would be necessary in this regard, where multi-level optimization algorithms proposed by Wang and Perera [57] can be pretty valuable. Machine learning algorithms can be quite beneficial in this regard, as reported by Dehghani [56] and Mojtaba et al. [58]. Such algorithms will help address complex design and operation problems.

#### 4.3 Impact of heat and cold waves on energy demand, generation, and transmission/distribution

As a consequence of climate change, the frequency of extreme hot and cold events has increased notably [1]. Therefore, a number of research studies are now focused on extreme hot and cold climate events. A significant increase in temperature is observed when considering the temperature variation at a finer temporal resolution, which increases the heating/cooling demand up to 70% [59]. Such a significant increase in the energy demand can easily influence thermal comfort within the building and the energy systems used to cater to the demand [60]. Accordingly, two different classes of papers were reviewed: (1) impact assessments considering energy demand and thermal comfort, and (2) impact assessments considering energy generation.

Quantifying the impacts of future climate variability and extreme climate events plays a vital role in improving climate resilience [61]. Therefore, a number of studies focused on evaluating the influence of climate change on energy demand, which can be further divided into two classes: studies based on historical data and future climate data. Publications on past climate data use past climate conditions to predict future energy demand while analyzing the trend. However, predicting the energy demand using past climate data does not fully capture the dynamics brought about by climate change [14]. Therefore, instead of depending on past climate data, most researchers use climate models to obtain the data [59,62,63] (Table 3). Regional and global climate models have been used in this context, taking a spectrum of representative concentration pathways (RCPs). Often, bottom-up models based on detailed hourly building simulations have been used to evaluate the impact of future climate variations on energy demand, such as in [59,62,63]. However, temperature-driven top-down approaches have also been used to quantify the impact of future climate variations on energy demand [50,64,65]. The majority of studies focused on assessing the impact of future climate variations on energy demand study the impact of extreme climate events on building energy demand (Table 3). Impacts on both heating and cooling have been considered in this context. In addition to assessing energy demand, Mutschler et al. [66], Tian et al. [67], and Lee & Levermore [68] assessed thermal comfort during extreme events. In addition to consider the energy demand Qin et al. [64] and Dasaraden et al. [69] considered the impact on renewable energy generation. Regional climate models used in the present state of the art provide mesoscale data. The majority of the present state-of-theart publications assessed the impact of future climate variations on the energy demand based on regional climate models (Table 3). Urban climate or urban microclimate models provide highresolution climate data, which are important when quantifying the impact of extreme events at the urban scale, since those models can represent the impact of the urban heat island, which can have a significant impact on energy infrastructure [70]. However, the majority of publications did not

consider the influence of urban climate when assessing the impact of future climate variations on energy demand (Table 3), and this can lead to a significant performance gap. Javanroodi and Nik [63] considered the influence of urban climate along with future climate conditions.

Quantifying the impact of extreme hot and cold events on distributed energy systems has been discussed quite comprehensively in the present state of the art (see Table 4). About six publications present either reviews or perspectives on this topic, summarizing the recent progress and promising research directions (see Table 4). The pool of papers can be classified into three classes: (1) design optimization, (2) optimal dispatch, and (3) resilience and reliability during extreme heat and cold events. Most of the publications, such as [71–74], focus on the optimal dispatch problem (directly or as a part of the design optimization). These studies focus on deriving the optimal operation strategy for the energy system to cater to the multi-energy demand (heating, cooling, and electricity). At the same time, several studies focused on optimizing the system design to cater to extreme energy demands while minimizing lifecycle cost. Different optimization techniques such as deterministic [72,75,76], stochastic [15,30,77], and robust [77] have been used in these studies. Uncertainty brought by future climate variations is considered in the stochastic models. In certain instances, hybrid approaches such as the stochastic-robust optimization technique have been used. Both low-probable high-impact (extremes) and high-probable low-impact events have been considered in these studies. The majority of the publications discuss the integration of renewable energy technologies while maintaining resilience during extreme events (Table 4). Although the majority of studies use detailed bottom-up approaches, several studies use the top-down approaches [76,78]. Several studies focus on the energy market and grid constraints along with the extreme conditions [76,78]. However, reasonable simplification has been made when considering the interactions between the different parties within the energy ecosystem and the uncertainties. Furthermore, climate flexibility has not been considered comprehensively in many studies.

	Modeling wildfire mpact on line rating thermal)	Renewables	Distributed generation	Storage	1	Optimization- operation	Optimal power flow		Optimal power shutoff problem		Future scenarios		Climate models	Grid Impact/ cascade failures		Transportation		Building		Stochastic modeling	Review	Instrumentation to detect wildfires
Sathaye et al. [33]		<u> </u>								٧		٧		٧								
Koufakis et al. [40]	v																					
Rhodes et al. [36]	٧		٧		٧		٧	٧														
Mohagheghi & Rebennack [43]	٧		٧		٧		٧	٧														
Trakas and Hatziargyriou [44]	٧		٧	٧	٧		٧	٧											٧			
Soulinaris et al. [79]	٧																					
Hay & Mohit [46]			1							v					v		v					
Hojjatinejad & Mona Ghassemi	v																				٧	
Erickson et al. [81]	٧																					
Wischkaemper et al. [42]																						٧
Muhs et al. [41]	V													٧								
Tandon et al. [45] Nazaripouya [39]	V		٧		٧		V	٧													٧	٧
Jazebi et al. [38]		v																				٧
Jazebi et al. [35]		٧																				v

### Table 1: Present state-of-the-art publications assessing the impact of wildfires on energy infrastructure

#### Table 2: Present state-of-the-art publications assessing the impact of hurricanes on energy infrastructure

	Network	Energy system	Demand	Dispatch	Optimal power flow	Design optimization	Stochastic model	Renewables	Storage	Dispatchable sources	Component level	Outage/cascade failure	Machine learning	Review/perspective
Shield et al. [49]	v											٧		
Bennett et al. [55]	v	v	v	v	٧	v		v	٧	٧				
Cole et al. [82]		v						v						
Tavakoli et al. [83]		v	v	v				v	٧					
Baghbanzadeh et al. [53]	v	v	v	v	٧					٧		v		
Parker et al. [51]														٧
Cicilio et al. [54]	v			v				v				٧		
Dehghani [56]	v	v	v				٧						٧	
Khomami & Sepasian [84]	v													
Pantua et al. [85]								v			٧			
Hashemi et al. [86]								v			٧			
Mojtaba et al. [58]	v	v	v	v				v	٧				٧	
Qian et al. [87]	v	v	v	v				v	٧	٧				
Chen et al. [52]														٧
Amirioun [88]	٧	٧		٧					٧	٧				
Rose et al. [89]								v			٧			
Hosseini & Parvania [90]	٧	٧		٧			٧		٧	٧			٧	

	Climate scenarios	Extreme events	GCM/RCM	Urban micro climate	Bottom-up models	Top- down	Electricity	Heating	Cooling	Thermal comfort	Renewable energy
Qin et al. [64]	v		v			٧	٧				٧
Mutschler et al. [66]	V		v		v	v		v	v		
Kalvelage [91]	V		v		v			v	v	v	
Yuchen et al. [62]	v	٧	v		v			v	v		
Larsen et al. [92]	V	٧				v					
Li et al. [93]		٧			v			v	v		
Hosseini et al. [94]	V		v		v			v	v		
Morakinyo et al. [65]		٧				v	v	v	v		
Crawley [95]	V	٧	v		v		V	v	v		
Tian et al. [67]	V		v		v		V	v	v	v	
Baniassadi [96]	v	٧	v		v		v	v	v		
Farah et al. [97]		٧			v			v	v		
Dasaraden et al. [69]	V				v		V	v	v	v	v
Lee & Levermore [68]	v		v		v				v		
Javanroodi & Nik [63]	v	٧	v	V	v			v	v		
Moazami et al. [59]	v	٧	V		v			v	v		

Table 3: Present state-of-the-art publications assessing the impact of heat waves on energy demand

	Climate scenarios	Extreme events	High probable low impact	GCM/RCM	Bottom-up models	Top-down	Electricity	Heating	Cooling	Wind	PV	Grid	Uncertainty	Resilience	Flexibility	Deterministic	Stochastic	Robust	Market	Design	Operation	Perspective/ review
Perera et al. [30]	٧			٧	٧		٧	٧	٧	٧	٧		٧		٧		٧			٧	٧	
Perera et al. [15]	٧	٧	٧	٧	v		٧	٧	٧	٧	٧		٧				٧	1		٧	٧	
Perera et al. [98]	٧	٧		٧	v		٧	٧	٧	٧	٧					٧				٧	٧	
Abdin et al. [75]	٧	٧		٧	v		٧			٧	٧					٧				٧	٧	
Wadsack et al. [72]		٧		٧	v		٧									٧					٧	
Pes et al. [99]		٧					٧			٧												
Su et al. [71]		٧					٧			٧	٧		٧			٧					٧	
Zhang et al. [100]		٧					٧			٧												
Ratnam et al. [101]							٧			٧	٧			٧	٧							٧
Nik et al. [14]	٧	٧		٧	v	٧	٧	٧	٧	٧	٧	v	٧				٧	v		٧	٧	٧
Chandramowli & Felder [78]	٧	٧		٧	٧	٧	٧	٧	٧	٧	٧	٧	٧						٧	٧	٧	٧
Ciscar & Dowling [76]	٧						٧	٧	٧	٧	٧					v			٧	٧		٧
Thomas et al. [102]	٧	٧		٧			٧			٧												
Wiel et al. [103]	٧	٧		٧			٧			٧	٧											
Martin & Rice [104]		٧					٧															
Demissie & Solomon [73]		٧					٧														٧	
Jordaan [105]	٧	٧			v		٧			٧	٧			٧								٧
Donk [106]	٧	٧		٧			٧			٧												
Huang et al. [107]	٧	٧		٧			٧			٧	٧											
Ward [108]	٧	٧					٧	٧	٧	٧	٧											٧
Höltinger et al. [74]		٧			v		٧			٧	٧		٧								٧	
Cross et al. [109]		٧					٧			٧												
Patt et al. [110]		٧					٧				٧											٧
Mavromatidis et al. [77]	٧			٧	٧		٧	٧	٧		٧							٧		٧	٧	
Mavromatidis et al. [111]	v		٧	٧	٧		٧	٧	٧		٧						v			٧	٧	

Table 4: Present state-of-the-art publications assessing the impact of heat waves on generation and distribution/transmission

#### Table 5: Recent studies accessing the impact of wildfires on buildings



#### 4.4 Limitations in the present state of the models

Significant growth in the research studies that focused on climate resilience considering various aspects of the energy ecosystem was observed in the present literature. The majority of these studies were based on bottom-up approaches that consider the detailed physics of the problem. Dispatch problems, optimal power flow problems, and energy system sizing problems, which have been already well defined in the present state of the art, have been extended to consider extreme climate events. Furthermore, uncertainties brought by future climate variation have also been captured, especially for models that consider extreme heat and cold events. Therefore, it can be concluded that present state-of-the-art models can be adopted to help understand and quantify the impacts caused by the four types of extreme events discussed.

#### • Considering the future climate data

Improving resilience of the energy infrastructure demands that we move beyond quantifying the impacts of extreme events. It is important to predict future events that may arise, derive strategies to operate existing infrastructure during such events, and improve the design of the energy ecosystem to improve resilience. Except for the extreme cold/heat events, the connectivity between energy and climate models was weak. Most of the assessments were conducted using data obtained for past events. Only two studies used future climate data in improving the resilience for wildfires, and one study considered future intensification of hurricanes due to climate change in the assessment. The work on the impact of wildfires on building energy is also limited (Table 5). As we come across more and more intensive and frequent extreme events, the resilience and reliability of infrastructure designed using historical data may be inadequate. Therefore, it is vital to link the energy models with future climate models.

#### • Considering uncertainties and use of stochastic models

Capturing uncertainties play a vital role during the decision-making process. Uncertainties brought about by climate, energy markets, the evolution and adoption of technologies, grid operation, and other factors play a key role. A number of studies have focused on capturing the uncertainties during energy system operation (dispatch), optimal power flow, energy system optimization, and grid expansion problems. Nonetheless, most of the studies on extreme events focused on deterministic models. Stochastic, robust, and hybrid (stochastic-robust) models have been used to model extreme hot/cold events on the energy infrastructure. Only a few studies used stochastic models to examine the impact of wildfires and hurricanes. Consideration of uncertainties brought by climate models and the impact of the human activities (consideration of uncertainties for the RCP or SSP [Shared Socioeconomic Pathway] taken) has been trivial. However, considering these uncertainties increases the complexity of the existing models, which makes it difficult to consider them within the modeling framework. Statistical and machine learning methods are promising ways to address these limitations, and these are gradually becoming popular.

#### • Planning for a resilient energy ecosystem

Most of the present state-of-the-art studies have focused on impact assessment. The impact of past events and forecasted future extreme events have been considered in these studies. A number of studies moved beyond the impact assessment and developed methodologies for optimal and resilient operation during extreme events (optimal power flow and optimal dispatch). Developing such methodologies help to retard the propagation of disruption and improve resource efficiency. However, the operation of energy infrastructure solely and heavily depends on the infrastructure, i.e., the superstructure of the grid, capacity, location, and type of energy sources and storage. Therefore, improving the design plays a vital role for improving climate resilience. However, the focus on design optimization has been quite limited when considering the present state of the art. For example, the authors did not find any publication about long-term energy planning to improve climate resilience for wildfires. Moreover, it is important to consider the resilience for multiple extreme events, not just a single event, especially during extreme events. In certain instances, the compound impact of multiple extreme events must be considered during the design phase. For example, in California, the energy infrastructure needs to be resilient to both extreme heat events and wildfires at the same time. Therefore, improving the resilience of the energy ecosystem for multiple extreme events and considering the compound impact is quite important. However, the authors did not come across such holistic design procedures to enhance the climate resilience of energy ecosystems. Many important research gaps must be addressed within the design improvement in energy infrastructure to improve energy ecosystem resilience.

#### 5. Propagation of disruptions

The main challenge of a virus following infection is its spread. The virus uses the human cell to multiply the virus population and continue the spread. In the case of COVID 19, the virus initially infects the respiratory tract and subsequently spreads through the respiratory system. Similarly, extreme events can obstruct a part of the energy ecosystem and spread through the entire ecosystem. Energy infrastructure is critical infrastructure that further increases the risk for such cascade failures. Such failures could cause much more damage than the local disruptions discussed in Section 4. However, the disruption can move forward beyond the limits of a single sector, as we observe for COVID-19. The virus can start its infection in the respiratory tract and propagate through the respiratory system to reach a level that it moves beyond the limits of the respiratory system. Thereafter, it starts to attack other systems, damaging the functionality of the heart, kidney, and digestive systems, further increasing the damage, as shown in Fig. 4.

Similarly, the disruption propagated within a single sector such as electricity could reach a critical limit where it moves beyond the boundary of electricity and starts to disrupt other interconnected infrastructure such as communications, gas, water, transportation, and others. Such propagation damaging multiple sectors could lead to a significant economic loss, disrupting social activities. Such propagation of disruption due to extreme events within the energy ecosystem considering single and multiple sectors are discussed in this section, respectively in Section 5.1 and Section 5.2. Possible ways to improve the resilience in the urban energy ecosystem holistically is addressed in Section 5.3.

#### 5.1 Propagation of disruptions in a single sector

Conducting vulnerability assessment for critical infrastructure has been a rich area of study, specifically for the electricity sector. Disruption in the energy infrastructure can occur due to several

reasons, including the extreme events discussed previously in Section 4. Following such an extreme event, the disruption can easily propagate and disrupt a large area, multiplying the impact of the extreme event by orders of magnitude. In general, vulnerability assessment of critical networked infrastructure has been a rich area of study. We do not intend to review the existing pool of studies on vulnerability assessment since it is not the focus of the study. However, this section briefly reviews a vulnerability assessment of the energy ecosystem solely related to extreme climate events. It is summarized in Table 6.

The present state-of-the-art literature on vulnerability assessment of energy infrastructure (mainly electricity) can be divided into two classes: (1) methods based on statistical approaches and (2) methods based on graph theory. Although both approaches have been used within the energy domain, the majority of the publications linked to climate events have used graph theory. The methods based on graph theory can be further classified into three groups: (1) simple graph, (2) weighted graph, and (3) detailed physics-based models considering the alternating current (AC) power flow. Vulnerability assessments have been linked to wildfires in many studies, as discussed in Secton 4.1 (see Table 1 for the summary). However, the impact of other extreme events has been fewer studies, as shown in Table 6. Vulnerability assessment considering multiple events is more challenging. Xu et al. [115] and Noebels et al. [116] conducted vulnerability assessments considering multiple extreme climate events. Furthermore, a majority of the publications are focused on electricity grids with dispatchable sources. The integration of renewables and storage has been less discussed in the present state of the art. Uncertainties of renewable energy generation and direct impacts on renewable energy generation during extreme events (especially for wind power generation during hurricanes) complicate consideration of renewable energy integration. The research performed by Athari & Wang [117], Noebels et al. [116], and Nesti et al. [118] are examples of vulnerability assessments that consider renewable energy technologies while also taking into account extreme events. Operational optimization (optimal dispatch and power flow) to minimize the tendency for cascade failures has been assessed in several studies. However, we did not come across any study that presents design improvements (optimal system sizing or optimal grid reinforcement) to avoid cascade failures, which will be an interesting future research direction.

#### 5.2 Propagation of disruptions in multiple sectors

Extreme events can lead to obstruct the operation of a certain part of the energy ecosystem. Being a critical infrastructure vulnerable to cascade failures, the disruption could propagate and disrupt the sector in a large area, as discussed in Section 5.1. Interdependency between different infrastructures could easily lead to obstruction of the connected infrastructure during extreme weather events. As a result, the propagation of the disruption will be accelerated and continue in more than one sector.

Let's take an example based on the communication and electricity networks. The communication network depends on the electricity network in order to obtain power for the operation. At the same time, the electricity network depends on the communication network to maintain communication between different parts of the electricity grid. Cascade failure in the electricity grid can collapse the operation of the communication network and lead to a cascade failure in communication networks. Failure in the communication network disrupts the control in the electricity grid, further extending the disruption. As a result, the obstruction taking place at a location can easily penetrate into a region and multiple sectors due to the interdependence in the infrastructure. Therefore, it is vital to improve the resilience of the interconnected infrastructure.

Understanding the propagation of disruption in interconnected infrastructure is much more complex than assessing the vulnerability of one sector [101]. As a result, literature that assesses the

vulnerability of interconnected networks is quite limited [13]. The majority of the publications on vulnerability assessment of interconnected networks related to climate are based on bottom-up approaches, since it is difficult to consider the interactions between different sectors when using statistical approaches (Table 7). Most of these studies are focused on coupled infrastructure where interdependencies between all the sectors are considered. For example when considering the electricity and communication networks (the same example taken before), rather than limiting the focus to the dependency of the communication network on the electricity network, they also consider the impact of the dependency of the electricity network on the communication network [119–121]. Some of these interactions are solely based on physical activities such as gas and electricity [122] networks, while some are cyber-physical interactions (for example communication and electricity grid) [122–124]. Several concepts have been introduced to handle the complexity that arises through the interactions between the infrastructures. The concept of multiplex networks was used initially to consider the interactions between different sectors [119,125,126]. However, conducting vulnerability assessment by using the concept of multiple networks is quite challenging. Therefore, Thacker et al. [120] proposed a system of system architecture to consider the complexity. Agent based modeling also has been used in the vulnerability assessment [127]. Later, it was further extended, and Shekhtman et al. [128] proposed a networks of network architecture to consider the interactions between different sectors. Although different approaches have been proposed to assess the vulnerability of interconnected infrastructure, research studies that use these different approaches for extreme climate events are quite limited. Intercomparison between different approaches cannot be observed when reviewing the literature. Therefore, it is difficult to propose the most promising approach when considering extreme climate events. This challenge led to many open research problems concerning the vulnerability assessment of interconnected infrastructure. It is essential to address these research gaps to improve the climate resilience of urban energy ecosystems.

	Paper	Statistical	Bottom-up	Stochastic	Optimization	Graph	Weighted graph	AC power flow	Vulnerability for cascade	Renewables	Storage	Dispatchable generators	Hurricanes	Extreme heat	Case study
Panigrahi & Maity [12	9]		٧					v	٧			٧			٧
Rios et al. [130]			٧	٧					v			v			
Qi [131]									٧			v			
Fang et al. [132]			٧		٧	v			٧			v			٧
Nakarmi [133]			٧		٧				٧						
Athari & Wang [117]				٧		٧		v	٧	٧	V	v			
Xu et al. [115]			٧	٧	٧	٧	v		v				v	٧	
Noebels et al. [134]			٧		٧	٧		v	v			V			
Wu et al. [135]						٧		v	v			V			
Noebels et al. [116]									v	٧	v		v	٧	٧
Gupta et al. [136]			٧	٧		٧			v						
Hu & Fan [137]			٧	٧		٧			v						
Eisenberg et al. [138]			٧	٧		٧			v						
Nesti et al. [118]		٧				٧			v	٧					
Fang et al. [139]			v		٧	٧			v			v			

Table 6: Research studies focused on vulnerability assessment of critical infrastructure for cascade failures taking into account a single sector

	Statistical	Bottom-up	Optimization	Graph	Weighted betweenness	Vulnerability for cascade	Flood	Climate	Extreme hot/cold	Coupled systems	Comm	Elec	Water	Cyber Physical system	Gas	Transport	Multiplex	System of systems	Network of networks	ABM	Other	Perspective
Jin et al. [119]		٧		٧	٧	٧				٧	٧	٧		v			٧					
Ouvang [122]		٧		٧	٧	v				٧		٧			v		٧					
Thacker et al.		v		٧	٧	v				v		v										
[120]											v		٧	٧		v		v				
Heracleous et al. [127]		v			V	V				٧										٧		
Kong [121]		v	v		v	v				v	v	v		v			٧					
Kamissoko et		٧								٧		٧				٧						
al. [140]						٧															v	
Guan and Chen [125]	٧	V		V	V	V				V		V				V	٧					
Yang et al. [141]		V		٧	V	٧				٧	٧	v									٧	
Bloomfield et al. [123]		v		v		v				٧	٧	v									٧	
Tsavdaroglou et al [124]		v					٧	v	v		٧	v	٧			٧					٧	
Arrighi et al.		v					v	v					٧			v						
[142]						v																
Shekhtman et																						
al. [128] Korkali at al			./	./								./							v			v
[126]		v	v	v		٧				v	v	v					٧					

Table 7: Research studies focused on vulnerability assessment of critical infrastructure for cascade failures taking interconnected energy infrastructure

	Quantitative	Qualitative	Statistical	Energy	Economic	Social	Water	Transportation	Climate	Networks	System models	Design optimization	Uncertainty	Renewables	Storage	Dispatchable	Grid	Cyber resilience	Perspectives /reviews	Urban resilience
Mazur et al. [143]		٧		٧	٧	٧														
		٧		٧													٧	٧	v	
Arghandeh et al. [144]	N			N			N												N	
Sukhwani et al. [145]	v			v			v												v	
Charifi & Vamagata [146]	v			٧	٧	٧	٧												v	
Sildriff & Falliagata [140]		٧		v	v	٧		v		v								v	v	
Mendizabal et al. [147]																				
Cantelmi [148]	v	٧		v	۷	۷		v		v									v	
	٧	٧		٧	٧	٧		٧											v	
Meerow et al. [149]		v			v	v	٧													V
Shi et al. [32]		•			•	•														٧
Ribeiro and Goncalves [150]		٧			٧	٧													v	V
					٧	٧													v	٧
Nelson et al. [151]	V																			.,
Cariolet [152]	v	v			v	v	v	v											v	v
	V	٧			٧	٧	٧	٧											٧	٧
Rus et al. [153]	v	v			v	v	v	v											v	v
Salimi & Ghamdi [154]		•		٧	•	•	•	•	٧										•	•

Table 8: Review publications on holistic assessment of urban energy resilience

#### 6. Broader understanding of the resilience

Section 5 clearly presents the complexity brought by the propagation of disruptions due to extreme climate events. Therefore, it is important to consider the concept of resilience broadly, rather than being limited to the boundaries of energy systems. Most of the studies on resilience were focused on the urban scale, using more holistic representation of resilience and taking into account the aspects of social, economic, technological, and environmental (natural systems) resilience [143,144,152]. Energy resilience is often considered under technological resilience. A more holistic (top-down approach) is used in this context. In contrast, resilience of energy systems has been broadly discussed with the energy community, where a top-down approach is used [144]. This section provides a discussion of both approaches, which can help to improve the understanding of the resilience concept being linked to the energy ecosystem.

The concepts of urban resilience or resilient cities have been broadly discussed in the present state of the art [152]. It has been studied from different perspectives, including sociological, economical, technological, and health. Later, a multidisciplinary approach combining all these different aspects was introduced to address resilience in a more descriptive manner [146–148]. Significant progress has been made since, covering a large number of publications, which are difficult to review one by one (Fig. 7). Similarly, a large pool of review papers cover urban resilience. We selected a group of review papers among them and analyzed the content discussed in these publications. The results are presented in Table 8.

The number of publications in Table 8 clearly shows that urban resilience has been discussed quite comprehensively, as well as broadly, although that discussion has been limited to one stream. Social, economic, and technological streams have been covered in many studies reviewed in these papers. However, considering social or economic aspects broadly requires consideration of a large number of parameters which increases the complexity of the analysis. Therefore, most of the publications select one specific parameter to present an entire stream (e.g., one parameter to represent economic resilience, another parameter to represent social resilience). More importantly, often finer temporal and spatial resolution is not used. Therefore, detailed system dynamic models have been implemented. However, it is challenging to directly extend the approach to the energy domain. Most of the studies have discussed energy resilience at the urban context more holistically. Therefore, energy resilience models. By doing so, a more comprehensive understanding of urban resilience can be obtained beyond a mere holistic understanding of urban resilience.





#### 7. Limitations and future perspectives

Improving the resilience and impact assessment for extreme events has gradually been getting more attention. When reviewing the progress in the present state of the art, it has been quite clear that there are many bottlenecks in quantifying the direct impact of extreme climate events, assessing the vulnerability of the energy ecosystem for extreme climate events, and improving resilience.

#### 7.1 Paradigm shift in the modeling architecture

Considering the influence of extreme events requires extending the existing models, which makes them quite bulky and computationally exhaustive. As a result, most of them are very specific and often can only be used to consider a single extreme event, rather than the compound impact. At the same time, most of the studies were limited to impact assessment; possible ways to improve the system operation and design have not been addressed. Therefore, there is considerable room to improve the existing models. First, existing energy models are becoming extremely bulky when considering future climate variations, especially at the urban scale. The present models often use a centralized approach when operating or designing energy infrastructure. However, there is a recent trend to move from the centralized approach to a decentralized approach, especially when operating or performing design changes in the energy ecosystem. Such an approach to move into decentralized models was not witnessed in the literature review. There are several challenges when implementing completely decentralized modeling architecture, since energy infrastructure depends heavily on spatial interactions, which are somewhat challenging to represent using completely decentralized architecture. Therefore, agent-based models based on leader-follower methods and hierarchicaldistributed models are becoming popular. This has led to a paradigm shift in the modeling architecture, moving from system models to the system of system models. Using concepts such as Urban Cell (Perera et al. 2021) based on the system of system architecture would be a beneficial approach.

#### 7.2 Use of machine learning techniques

The energy systems domain has been quite successful, based on the bottom-up approaches with analytical models. Conversely, there were many limitations of using top-down or statistical approaches. Mainly, performance indicators of the energy system models were based on the detailed system dynamics. Statistical models fail to present such detail system dynamics. Furthermore, topdown models are based on poor spatial and temporal resolution, making them difficult to use for operational purposes. Therefore, statistical models cannot be directly used to address the modeling challenges presented by extreme climate events. Machine learning algorithms have become quite attractive in this context, and have proven to be efficient when handling complexity. Graph neural networks have already been used to investigate the vulnerability for cascade failures during extreme climate events and have shown the potential to be competitive with bottom-up graph models. More importantly, machine learning models are data-driven, and have already shown potential to be efficiently linked with climate models. Machine learning techniques such as reinforcement learning techniques have been deployed efficiently for dispatch and optimal power flow problems where energy system operation considering complex settings has been captured. Furthermore, reinforcement learning and transfer learning have been used with energy system design. Therefore, machine learning techniques have shown the potential for further investigating usability to improve the climate resilience of energy infrastructure. Therefore, introducing machine learning techniques to enhance the climate resilience of energy ecosystems will be an interesting research direction.

#### 7.3 Enhancing the interlinks between energy, climate, and human systems models

Understanding the impact of future climate variations on the energy infrastructure requires a detailed understanding of the interactions between climate, energy, and human systems. Future climate variations notably influence the resilience of the urban energy ecosystem. Already, there are many research gaps in capturing the impact of future climate variations when developing energy models, as highlighted in Section 4.4. Therefore, further improving the interconnectivity between energy and climate system models plays a vital role (Fig. 8). Similarly, human systems play a vital role in considering both climate and energy. The climate system already considers the impact of the human system (socio-economic) in the long run through the integrated assessment models. Similarly, the human system influences the energy sector (both design and operation of energy infrastructure). For example, changes in socio-economic factors (taken within the human system) may significantly influence design solutions and the transition path taken by the energy infrastructure. At the same time, the impact of human systems would be quite significant in relation to the operation of energy infrastructure during extreme events. For example, heating and cooling demand will be significantly influenced during extreme cold and hot events. Similarly, equipment usage patterns could notably change, which will affect energy demand. Considering the impact of human systems is important when improving the resilience for extreme climate events. Therefore, improving the interconnection between energy, humans, and climate systems plays a major role. Handling the complexity of the models when considering these interactions presents a major challenge. Paradigm change in the existing modeling frameworks, as explained in 7.1 and 7.2, will be crucial.



Fig. 8 moving from (a) a disintegrated to (b) an integrated modeling framework to improve the interconnectivity in models for enhancing resilience of urban energy ecosystem

#### 8. Conclusions

Improving the resilience of the urban energy ecosystem is vital, although the definition of resilience has been quite vague, especially within the energy system domain. The focus of these definitions varies from improving the resilience of a single component to the entire social system in a city. Some of the models are extremely qualitative with a very broad scope, while the others are quantitative. The same problem has been observed when considering the climate resilience of the energy ecosystem.

A comprehensive literature review is performed considering the threat caused by wildfires, hurricanes, and extreme hot and cold events brought up by future climate variations on the energy ecosystem. Research studies have focused mainly on minimizing the direct impact of extreme events. During the review process, two clear strategies were observed in the present state of the art to improve the resilience of the energy ecosystem:

1. Short-term planning: Develop resilient operational strategies to cope with extreme events

#### 2. Long-term planning: Improve the design to withstand extreme events

Most of the publications have focused on minimizing the direct impact of extreme events by improving short-term planning strategies such as optimal dispatch and optimal power flow. More than 90% of studies focused on a single extreme event, and improving resilience for multiple extreme events was not considered.

#### • Main bottlenecks in the present state of the art

There were several major bottlenecks observed in the present state of the art which hinder the process of improving the resilience of urban energy ecosystem.

#### A clear research gap was observed in developing tools for long-term planning.

Most of the modeling tools are used to operate existing energy infrastructure during an extreme event with a minimal impact on the day-to-day activities. However, besides operating the existing energy infrastructure, it is important to design energy infrastructure in a manner that they can withstand extreme events with a minimal impact on social activities. Towards this end, the present state of the art models need to be updated to evaluate long-term planning activities.

#### Future climate data

Energy community often relies on the past weather data. Except for extreme hot/cold events, most of the studies do not consider future climate data. However, relying on the past data does not guarantee the resilience of the energy ecosystem for future extreme events since the climate change is intensifying the severity of extreme climate events. Therefore, many interesting research problems arise when trying to improve the resilience for wildfires and hurricanes considering future climate data.

#### Uncertainties and stochastic models

Considering uncertainties plays a vital role in enhancing the resilience especially during the long-term planning process. One major challenge in this context is consideration of future climate data. Climate models, which have relatively poor temporal and spatial resolution, result in many uncertainties. At the same time there is a requirement to consider more than one climate model in order to enhance the accuracy. Furthermore, technology maturity and market conditions bring many uncertainties that need to be considered in planning process. Although there are a few models that consider these uncertainties, majority of the publications do not consider the uncertainties. Moving into stochastic models that consider the uncertainty may notably increase the complexity that these models cannot handle.

#### Facing multiple extreme events

Most of the recent studies focus on a single extreme event. As tabulated in Table 1-5 these studies either focus on wildfires, extreme hot/cold event, or hurricane. However, we need to understand that there can be a possibility that energy ecosystem needs to withstand multiple extreme events at the same time. For example, wildfires taking place during the extreme hot period require the energy ecosystem to withstand both these events at the same time. The present modeling tools are not capable of handling such simultaneous multiple extreme events.

#### Propagation cascade failures beyond the electricity sector

Vulnerability assessment for cascade failures, especially in the electricity sector, plays a major role in improving resilience. There has been reasonable progress in assessing the n-1 security and

vulnerability of wildfires addressing the optimal power shutdown problem. Although such vulnerability assessment in the electricity sector has been quite common, studies that focus on extreme climate events have been fewer. More importantly, vulnerability assessments considering interconnected energy infrastructure have been quite limited. The impact of extreme events on coupled cyber-physical systems such as electricity- communication have been performed. However, the resilience of such interconnected infrastructure within the energy ecosystem for extreme climate events has not been performed yet.

There are several promising approaches to address these bottlenecks. We show that obtaining a more holistic understanding of the vulnerability of the energy infrastructure to extreme climate events could play a crucial role in addressing some of aforementioned limitations.

#### • Promising approaches

#### Covid analogy

the Covid analogy was developed to better understand which critical aspects need to be focused on. The three main stages of the coronavirus spread in human bodies were used to explain the propagation of disruption in the energy ecosystem and ways to improve resilience in different parts of the energy ecosystem.

#### Shift in the modeling architecture

In general, extreme climate events extend the typical energy system models used in the present state of the art, making them bulky and more complex. Therefore, improving the modeling framework is essential in order to improve the climate resilience of energy infrastructure. Moving from the centralized modeling architecture towards a modular-based decentralized architecture could be quite promising in this context, and could help to handle the increasing complexity. Methods such as Urban Cell could be helpful in this regard. Modeling architectures such as system-of-systems and networkof-networks can be quite interesting in this context.

#### Machine learning techniques

Machine learning methods that can help to improve the resilience of the energy ecosystem are also becoming popular within the energy system community. These methods can help to improve the connectivity between energy system models and climate and human system models.

#### Integrated assessment

In order to quantify the impact of future climate variations on the energy infrastructure a detailed understanding of the interactions between climate, energy, and human systems is essential. There are many research gaps in capturing the impact of future climate variations when developing energy models. However, there are more research gaps when linking human systems with both energy and climate systems. Therefore, it is essential to develop integrated assessment models coupling human, climate and energy systems.

#### Acknowledgments

This work was supported by the Laboratory Directed Research and Development (LDRD) Program of Lawrence Berkeley National Laboratory, provided by the Director, Office of Science, of the U.S. Department of Energy under Contract No. DE-AC02-05CH11231.

#### References

- [1] IPCC Fifth Assessment Synthesis Report. IPCC 5th Assess Synth Rep n.d. http://ar5-syr.ipcc.ch/ (accessed May 25, 2018).
- [2] Zhou Y, Varquez ACG, Kanda M. High-resolution global urban growth projection based on multiple applications of the SLEUTH urban growth model. Sci Data 2019;6:34. https://doi.org/10.1038/s41597-019-0048-z.
- [3] World Bank. Cities and Climate Change : An Urgent Agenda. Washington, DC: World Bank; 2010. https://doi.org/10/12/14981086/cities-climate-change-urgent-agenda.
- [4] Mohajeri N, Perera ATD, Coccolo S, Mosca L, Le Guen M, Scartezzini J-L. Integrating urban form and distributed energy systems: Assessment of sustainable development scenarios for a Swiss village to 2050. Renew Energy 2019;143:810–26. https://doi.org/10.1016/j.renene.2019.05.033.
- [5] Guen ML, Mosca L, Perera ATD, Coccolo S, Mohajeri N, Scartezzini J-L. Improving the energy sustainability of a Swiss village through building renovation and renewable energy integration. Energy Build 2018;158:906–23. https://doi.org/10.1016/j.enbuild.2017.10.057.
- [6] Davis SJ, Lewis NS, Shaner M, Aggarwal S, Arent D, Azevedo IL, et al. Net-zero emissions energy systems. Science 2018;360. https://doi.org/10.1126/science.aas9793.
- [7] Davis SJ, Caldeira K, Matthews HD. Future CO2 Emissions and Climate Change from Existing Energy Infrastructure. Science 2010;329:1330–3. https://doi.org/10.1126/science.1188566.
- [8] Ramsebner J, Haas R, Ajanovic A, Wietschel M. The sector coupling concept: A critical review. WIREs Energy Environ n.d.;n/a:e396. https://doi.org/10.1002/wene.396.
- [9] Perera ATD, Javanroodi K, Wang Y, Hong T. Urban cells: Extending the energy hub concept to facilitate sector and spatial coupling. Adv Appl Energy 2021;3:100046. https://doi.org/10.1016/j.adapen.2021.100046.
- [10] Mai TT, Jadun P, Logan JS, McMillan CA, Muratori M, Steinberg DC, et al. Electrification Futures Study: Scenarios of Electric Technology Adoption and Power Consumption for the United States. 2018. https://doi.org/10.2172/1459351.
- [11] Murray P, Orehounig K, Grosspietsch D, Carmeliet J. A comparison of storage systems in neighbourhood decentralized energy system applications from 2015 to 2050. Appl Energy 2018;231:1285–306. https://doi.org/10.1016/j.apenergy.2018.08.106.
- [12] Ram M, Osorio-Aravena JC, Aghahosseini A, Bogdanov D, Breyer C. Job creation during a climate compliant global energy transition across the power, heat, transport, and desalination sectors by 2050. Energy 2022;238:121690. https://doi.org/10.1016/j.energy.2021.121690.
- [13] Panteli M, Mancarella P. Influence of extreme weather and climate change on the resilience of power systems: Impacts and possible mitigation strategies. Electr Power Syst Res 2015;127:259–70. https://doi.org/10.1016/j.epsr.2015.06.012.
- [14] Nik VM, Perera ATD, Chen D. Towards climate resilient urban energy systems: a review. Natl Sci Rev 2021;8. https://doi.org/10.1093/nsr/nwaa134.
- [15] Perera ATD, Nik VM, Chen D, Scartezzini J-L, Hong T. Quantifying the impacts of climate change and extreme climate events on energy systems. Nat Energy 2020;5:150–9. https://doi.org/10.1038/s41560-020-0558-0.
- [16] Busby JW, Baker K, Bazilian MD, Gilbert AQ, Grubert E, Rai V, et al. Cascading risks: Understanding the 2021 winter blackout in Texas. Energy Res Soc Sci 2021;77:102106. https://doi.org/10.1016/j.erss.2021.102106.
- [17] Hauser C, Sandoval E. Death Toll From Texas Winter Storm Continues to Rise. N Y Times 2021.
- [18] Chicco G, Mancarella P. Distributed multi-generation: A comprehensive view. Renew Sustain Energy Rev 2009;13:535–51. https://doi.org/10.1016/j.rser.2007.11.014.

- [19] Zhang X, He M, Zhang Y. A review of research on the Kalina cycle. Renew Sustain Energy Rev 2012;16:5309–18. https://doi.org/10.1016/j.rser.2012.05.040.
- [20] Imran M, Haglind F, Asim M, Zeb Alvi J. Recent research trends in organic Rankine cycle technology: A bibliometric approach. Renew Sustain Energy Rev 2018;81:552–62. https://doi.org/10.1016/j.rser.2017.08.028.
- [21] Mirhassani S, Ong HC, Chong WT, Leong KY. Advances and challenges in grid tied photovoltaic systems. Renew Sustain Energy Rev 2015;49:121–31. https://doi.org/10.1016/j.rser.2015.04.064.
- [22] Agustín JLB, Lo' pez RD. Simulation and optimization of stand-alone hybrid renewable energy systems. Renew Sustain Energy Rev 2009;13:2111–8.
- [23] Bajpai P, Dash V. Hybrid renewable energy systems for power generation in stand-alone applications: A review. Renew Sustain Energy Rev 2012;16:2926–39. https://doi.org/10.1016/j.rser.2012.02.009.
- [24] Geidl M, Koeppel G, Favre-Perrod P, Klockl B, Andersson G, Frohlich K. Energy hubs for the future. IEEE Power Energy Mag 2007;5:24–30. https://doi.org/10.1109/MPAE.2007.264850.
- [25] Tsujimoto M, Kajikawa Y, Tomita J, Matsumoto Y. A review of the ecosystem concept Towards coherent ecosystem design. Technol Forecast Soc Change 2018;136:49–58. https://doi.org/10.1016/j.techfore.2017.06.032.
- [26] Hamilton W a. H. Resilience and the city: the water sector. Proc Inst Civ Eng Urban Des Plan 2009;162:109–21. https://doi.org/10.1680/udap.2009.162.3.109.
- [27] Coaffee J. Towards Next-Generation Urban Resilience in Planning Practice: From Securitization to Integrated Place Making. Plan Pract Res 2013;28:323–39. https://doi.org/10.1080/02697459.2013.787693.
- [28] Desouza KC, Flanery TH. Designing, planning, and managing resilient cities: A conceptual framework. Cities 2013;35:89–99. https://doi.org/10.1016/j.cities.2013.06.003.
- [29] Asprone D, Cavallaro M, Latora V, Manfredi G, Nicosia V. Urban network resilience analysis in case of earthquakes. In: Deodatis G, Ellingwood BR, Frangopol DM, editors. vol. Taylor&Francis, London: CRC Press; 2014, p. 4069–75.
- [30] Perera ATD, Nik VM, Wickramasinghe PU, Scartezzini J-L. Redefining energy system flexibility for distributed energy system design. Appl Energy 2019;253:113572. https://doi.org/10.1016/j.apenergy.2019.113572.
- [31] Perera ATD, Nik VM, Mauree D, Scartezzini J-L. An integrated approach to design site specific distributed electrical hubs combining optimization, multi-criterion assessment and decision making. Energy 2017;134:103–20. https://doi.org/10.1016/j.energy.2017.06.002.
- [32] Shi Y, Zhai G, Xu L, Zhou S, Lu Y, Liu H, et al. Assessment methods of urban system resilience: From the perspective of complex adaptive system theory. Cities 2021;112:103141. https://doi.org/10.1016/j.cities.2021.103141.
- [33] Sathaye JA, Dale LL, Larsen PH, Fitts GA, Koy K, Lewis SM, et al. Rising Temps, Tides, and Wildfires: Assessing the Risk to California's Energy Infrastructure from Projected Climate Change. IEEE Power Energy Mag 2013;11:32–45. https://doi.org/10.1109/MPE.2013.2245582.
- [34] Luo N, Weng W, Xu X, Hong T, Fu M, Sun K. Assessment of occupant-behavior-based indoor air quality and its impacts on human exposure risk: A case study based on the wildfires in Northern California. Sci Total Environ 2019;686:1251–61. https://doi.org/10.1016/j.scitotenv.2019.05.467.
- [35] Review of Wildfire Management Techniques—Part II: Urgent Call for Investment in Research and Development of Preventative Solutions n.d. https://ieeexplore.ieee.org/document/8767948 (accessed October 28, 2021).
- [36] Rhodes N, Ntaimo L, Roald L. Balancing Wildfire Risk and Power Outages Through Optimized Power Shut-Offs. IEEE Trans Power Syst 2021;36:3118–28. https://doi.org/10.1109/TPWRS.2020.3046796.

- [37] Romanello M, McGushin A, Napoli CD, Drummond P, Hughes N, Jamart L, et al. The 2021 report of the Lancet Countdown on health and climate change: code red for a healthy future. The Lancet 2021;398:1619–62. https://doi.org/10.1016/S0140-6736(21)01787-6.
- [38] Jazebi S, de León F, Nelson A. Review of Wildfire Management Techniques—Part I: Causes, Prevention, Detection, Suppression, and Data Analytics. IEEE Trans Power Deliv 2020;35:430– 9. https://doi.org/10.1109/TPWRD.2019.2930055.
- [39] Nazaripouya H. Power Grid Resilience under Wildfire: A Review on Challenges and Solutions.
  2020 IEEE Power Energy Soc. Gen. Meet. PESGM, 2020, p. 1–5.
  https://doi.org/10.1109/PESGM41954.2020.9281708.
- [40] Koufakis EI, Tsarabaris PT, Katsanis JS, Karagiannopoulos CG, Bourkas PD. A Wildfire Model for the Estimation of the Temperature Rise of an Overhead Line Conductor. IEEE Trans Power Deliv 2010;25:1077–82. https://doi.org/10.1109/TPWRD.2009.2035128.
- [41] Muhs J, Parvania M, Nguyen HT, Palmer JA. Characterizing Probability of Wildfire Ignition Caused by Power Distribution Lines. IEEE Trans Power Deliv 2020:1–1. https://doi.org/10.1109/TPWRD.2020.3047101.
- [42] Wischkaemper JA, Benner CL, Don Russell B, Muthu Manivannan K. Application of advanced electrical waveform monitoring and analytics for reduction of wildfire risk. ISGT 2014, 2014, p. 1–5. https://doi.org/10.1109/ISGT.2014.6816487.
- [43] Mohagheghi S, Rebennack S. Optimal resilient power grid operation during the course of a progressing wildfire. Int J Electr Power Energy Syst 2015;73:843–52. https://doi.org/10.1016/j.ijepes.2015.05.035.
- [44] Trakas DN, Hatziargyriou ND. Optimal Distribution System Operation for Enhancing Resilience Against Wildfires. IEEE Trans Power Syst 2018;33:2260–71. https://doi.org/10.1109/TPWRS.2017.2733224.
- [45] Tandon S, Grijalva S, Molzahn DK. Motivating the Use of Dynamic Line Ratings to Mitigate the Risk of Wildfire Ignition. 2021 IEEE Power Energy Conf. III. PECI, 2021, p. 1–7. https://doi.org/10.1109/PECI51586.2021.9435252.
- [46] Hay C, Chhabra M. The impact of wildfires and beneficial electrification on electricity rates in PG&E's service territory. Electr J 2020;33:106710. https://doi.org/10.1016/j.tej.2020.106710.
- [47] 2021 WILDFIRE MITIGATION PLAN REPORT 2021:1013.
- [48] Schoolov K. How PG&E is fighting its massive wildfire problem with microgrids, power shutoffs and cutting down trees. CNBC 2021. https://www.cnbc.com/2021/07/03/how-californias-pge-is-fighting-its-massive-wildfire-problem.html (accessed February 15, 2022).
- [49] Shield SA, Quiring SM, Pino JV, Buckstaff K. Major impacts of weather events on the electrical power delivery system in the United States. Energy 2021;218:119434. https://doi.org/10.1016/j.energy.2020.119434.
- [50] Larsen P, LaCommare K, Eto J, Sweeney J. Assessing Changes in the Reliability of the U.S. Electric Power System 2015.
- [51] Feldpausch-Parker AM, Peterson TR, Stephens JC, Wilson EJ. Smart grid electricity system planning and climate disruptions: A review of climate and energy discourse post-Superstorm Sandy. Renew Sustain Energy Rev 2018;82:1961–8. https://doi.org/10.1016/j.rser.2017.06.015.
- [52] Chen H, Bresler FS, Bryson ME, Seiler K, Monken J. Toward Bulk Power System Resilience: Approaches for Regional Transmission Operators. IEEE Power Energy Mag 2020;18:20–30. https://doi.org/10.1109/MPE.2020.2985437.
- [53] Baghbanzadeh D, Salehi J, Gazijahani FS, Shafie-khah M, Catalão JPS. Resilience improvement of multi-microgrid distribution networks using distributed generation. Sustain Energy Grids Netw 2021;27:100503. https://doi.org/10.1016/j.segan.2021.100503.
- [54] Cicilio P, Swartz L, Vaagensmith B, Rieger C, Gentle J, McJunkin T, et al. Electrical grid resilience framework with uncertainty. Electr Power Syst Res 2020;189:106801. https://doi.org/10.1016/j.epsr.2020.106801.

- [55] Bennett JA, Trevisan CN, DeCarolis JF, Ortiz-García C, Pérez-Lugo M, Etienne BT, et al. Extending energy system modelling to include extreme weather risks and application to hurricane events in Puerto Rico. Nat Energy 2021:1–10. https://doi.org/10.1038/s41560-020-00758-6.
- [56] Dehghani NL, Jeddi AB, Shafieezadeh A. Intelligent hurricane resilience enhancement of power distribution systems via deep reinforcement learning. Appl Energy 2021;285:116355. https://doi.org/10.1016/j.apenergy.2020.116355.
- [57] Wang Z, Perera ATD. Integrated platform to design robust energy internet. Appl Energy 2020;269:114942. https://doi.org/10.1016/j.apenergy.2020.114942.
- [58] Moradi-Sepahvand M, Amraee T, Sadeghi Gougheri S. Deep Learning-Based Hurricane Resilient Co-planning of Transmission Lines, Battery Energy Storages and Wind Farms. IEEE Trans Ind Inform 2021:1–1. https://doi.org/10.1109/TII.2021.3074397.
- [59] Moazami A, Nik VM, Carlucci S, Geving S. Impacts of future weather data typology on building energy performance – Investigating long-term patterns of climate change and extreme weather conditions. Appl Energy 2019;238:696–720. https://doi.org/10.1016/j.apenergy.2019.01.085.
- [60] Nik VM, Perera ATD. The Importance of Developing Climate-Resilient Pathways for Energy Transition and Climate Change Adaptation. One Earth 2020;3:423–4. https://doi.org/10.1016/j.oneear.2020.09.013.
- [61] Mauree D, Naboni E, Coccolo S, Perera ATD, Nik VM, Scartezzini J-L. A review of assessment methods for the urban environment and its energy sustainability to guarantee climate adaptation of future cities. Renew Sustain Energy Rev 2019;112:733–46. https://doi.org/10.1016/j.rser.2019.06.005.
- [62] Yang Y, Javanroodi K, Nik VM. Climate change and energy performance of European residential building stocks – A comprehensive impact assessment using climate big data from the coordinated regional climate downscaling experiment. Appl Energy 2021;298:117246. https://doi.org/10.1016/j.apenergy.2021.117246.
- [63] Javanroodi K, Nik VM. Impacts of Microclimate Conditions on the Energy Performance of Buildings in Urban Areas. Buildings 2019;9:189. https://doi.org/10.3390/buildings9080189.
- [64] Qin P, Xu H, Liu M, Xiao C, Forrest KE, Samuelsen S, et al. Assessing concurrent effects of climate change on hydropower supply, electricity demand, and greenhouse gas emissions in the Upper Yangtze River Basin of China. Appl Energy 2020;279:115694. https://doi.org/10.1016/j.apenergy.2020.115694.
- [65] Morakinyo TE, Ren C, Shi Y, Lau KK-L, Tong H-W, Choy C-W, et al. Estimates of the impact of extreme heat events on cooling energy demand in Hong Kong. Renew Energy 2019;142:73– 84. https://doi.org/10.1016/j.renene.2019.04.077.
- [66] Mutschler R, Rüdisüli M, Heer P, Eggimann S. Benchmarking cooling and heating energy demands considering climate change, population growth and cooling device uptake. Appl Energy 2021;288:116636. https://doi.org/10.1016/j.apenergy.2021.116636.
- [67] Tian Z, Zhang S, Deng J, Dorota Hrynyszyn B. Evaluation on Overheating Risk of a Typical Norwegian Residential Building under Future Extreme Weather Conditions. Energies 2020;13:658. https://doi.org/10.3390/en13030658.
- [68] Lee SE, Levermore GJ. Simulating urban heat island effects with climate change on a Manchester house. Build Serv Eng Res Technol 2013;34:203–21. https://doi.org/10.1177/0143624412439485.
- [69] Mauree D, Coccolo S, Perera A, Nik V, Scartezzini J-L, Naboni E, et al. A New Framework to Evaluate Urban Design Using Urban Microclimatic Modeling in Future Climatic Conditions. Sustainability 2018;10:1134. https://doi.org/10.3390/su10041134.
- [70] Perera ATD, Coccolo S, Scartezzini J-L, Mauree D. Quantifying the impact of urban climate by extending the boundaries of urban energy system modeling. Appl Energy 2018;222:847–60. https://doi.org/10.1016/j.apenergy.2018.04.004.

- [71] Su Y, Kern JD, Reed PM, Characklis GW. Compound hydrometeorological extremes across multiple timescales drive volatility in California electricity market prices and emissions. Appl Energy 2020;276:115541. https://doi.org/10.1016/j.apenergy.2020.115541.
- [72] Wadsack K, Acker TL. Climate change and future power systems: the importance of energy storage in reduced-hydropower systems in the American Southwest. Clean Energy 2019;3:241–50. https://doi.org/10.1093/ce/zkz018.
- [73] Demissie AA, Solomon AA. Power system sensitivity to extreme hydrological conditions as studied using an integrated reservoir and power system dispatch model, the case of Ethiopia. Appl Energy 2016;182:442–63. https://doi.org/10.1016/j.apenergy.2016.08.106.
- [74] Höltinger S, Mikovits C, Schmidt J, Baumgartner J, Arheimer B, Lindström G, et al. The impact of climatic extreme events on the feasibility of fully renewable power systems: A case study for Sweden. Energy 2019;178:695–713. https://doi.org/10.1016/j.energy.2019.04.128.
- [75] Abdin IF, Fang Y-P, Zio E. A modeling and optimization framework for power systems design with operational flexibility and resilience against extreme heat waves and drought events. Renew Sustain Energy Rev 2019;112:706–19. https://doi.org/10.1016/j.rser.2019.06.006.
- [76] Ciscar J-C, Dowling P. Integrated assessment of climate impacts and adaptation in the energy sector. Energy Econ 2014;46:531–8. https://doi.org/10.1016/j.eneco.2014.07.003.
- [77] Mavromatidis G, Orehounig K, Carmeliet J. Comparison of alternative decision-making criteria in a two-stage stochastic program for the design of distributed energy systems under uncertainty. Energy 2018;156:709–24. https://doi.org/10.1016/j.energy.2018.05.081.
- [78] Chandramowli SN, Felder FA. Impact of climate change on electricity systems and markets A review of models and forecasts. Sustain Energy Technol Assess 2014;5:62–74. https://doi.org/10.1016/j.seta.2013.11.003.
- [79] Soulinaris GK, Halevidis CD, Polykrati AD, Bourkas PD. Evaluation of the thermal stresses and dielectric phenomena in the investigation of the causes of wildfires involving distribution power lines. Electr Power Syst Res 2014;117:76–83. https://doi.org/10.1016/j.epsr.2014.07.031.
- [80] Hojjatinejad S, Ghassemi M. Resiliency Enhancement against Wildfires. 2021 IEEE Power Energy Soc. Innov. Smart Grid Technol. Conf. ISGT, 2021, p. 1–5. https://doi.org/10.1109/ISGT49243.2021.9372206.
- [81] Erickson E, Slobodin R, Poshtan M, Taufik T, Callenes J. Using Power Infrastructures for Wildfire Detection in California. 2020 IEEE Power Energy Soc. Innov. Smart Grid Technol. Conf. ISGT, 2020, p. 1–5. https://doi.org/10.1109/ISGT45199.2020.9087666.
- [82] Cole W, Greer D, Lamb K. The potential for using local PV to meet critical loads during hurricanes. Sol Energy 2020;205:37–43. https://doi.org/10.1016/j.solener.2020.04.094.
- [83] Tavakoli M, Shokridehaki F, Funsho Akorede M, Marzband M, Vechiu I, Pouresmaeil E. CVaRbased energy management scheme for optimal resilience and operational cost in commercial building microgrids. Int J Electr Power Energy Syst 2018;100:1–9. https://doi.org/10.1016/j.ijepes.2018.02.022.
- [84] Sadeghi Khomami M, Sepasian MS. Pre-hurricane optimal placement model of repair teams to improve distribution network resilience. Electr Power Syst Res 2018;165:1–8. https://doi.org/10.1016/j.epsr.2018.08.016.
- [85] Pantua CAJ, Calautit JK, Wu Y. A fluid-structure interaction (FSI) and energy generation modelling for roof mounted renewable energy installations in buildings for extreme weather and typhoon resilience. Renew Energy 2020;160:770–87. https://doi.org/10.1016/j.renene.2020.06.023.
- [86] Hashemi MR, Kresning B, Hashemi J, Ginis I. Assessment of hurricane generated loads on offshore wind farms; a closer look at most extreme historical hurricanes in New England. Renew Energy 2021;175:593–609. https://doi.org/10.1016/j.renene.2021.05.042.

- [87] Qian M, Chen N, Chen Y, Chen C, Qiu W, Zhao D, et al. Optimal Coordinated Dispatching Strategy of Multi-Sources Power System with Wind, Hydro and Thermal Power Based on CVaR in Typhoon Environment. Energies 2021;14:3735. https://doi.org/10.3390/en14133735.
- [88] Amirioun MH, Aminifar F, Shahidehpour M. Resilience-Promoting Proactive Scheduling Against Hurricanes in Multiple Energy Carrier Microgrids. IEEE Trans Power Syst 2019;34:2160–8. https://doi.org/10.1109/TPWRS.2018.2881954.
- [89] Rose S, Jaramillo P, Small MJ, Apt J. Quantifying the Hurricane Catastrophe Risk to Offshore Wind Power. Risk Anal 2013;33:2126–41. https://doi.org/10.1111/risa.12085.
- [90] Hosseini MM, Parvania M. Resilient Operation of Distribution Grids Using Deep Reinforcement Learning. IEEE Trans Ind Inform 2021:1–1. https://doi.org/10.1109/TII.2021.3086080.
- [91] Kalvelage K, Passe U, Rabideau S, Takle ES. Changing climate: The effects on energy demand and human comfort. Energy Build 2014;76:373–80. https://doi.org/10.1016/j.enbuild.2014.03.009.
- [92] Larsen MAD, Petrović S, Radoszynski AM, McKenna R, Balyk O. Climate change impacts on trends and extremes in future heating and cooling demands over Europe. Energy Build 2020;226:110397. https://doi.org/10.1016/j.enbuild.2020.110397.
- [93] Li M, Shi J, Guo J, Cao J, Niu J, Xiong M. Climate Impacts on Extreme Energy Consumption of Different Types of Buildings. PLOS ONE 2015;10:e0124413. https://doi.org/10.1371/journal.pone.0124413.
- [94] Hosseini M, Tardy F, Lee B. Cooling and heating energy performance of a building with a variety of roof designs; the effects of future weather data in a cold climate. J Build Eng 2018;17:107–14. https://doi.org/10.1016/j.jobe.2018.02.001.
- [95] Crawley DB. Estimating the impacts of climate change and urbanization on building performance. J Build Perform Simul 2008;1:91–115. https://doi.org/10.1080/19401490802182079.
- [96] Baniassadi A, Sailor DJ, Krayenhoff ES, Broadbent AM, Georgescu M. Passive survivability of buildings under changing urban climates across eight US cities. Environ Res Lett 2019;14:074028. https://doi.org/10.1088/1748-9326/ab28ba.
- [97] Farah S, Whaley D, Saman W, Boland J. Integrating climate change into meteorological weather data for building energy simulation. Energy Build 2019;183:749–60. https://doi.org/10.1016/j.enbuild.2018.11.045.
- [98] Perera ATD, Javanroodi K, Nik VM. Climate resilient interconnected infrastructure: Cooptimization of energy systems and urban morphology. Appl Energy 2021;285:116430. https://doi.org/10.1016/j.apenergy.2020.116430.
- [99] Pes MP, Pereira EB, Marengo JA, Martins FR, Heinemann D, Schmidt M. Climate trends on the extreme winds in Brazil. Renew Energy 2017;109:110–20. https://doi.org/10.1016/j.renene.2016.12.101.
- [100] Zhang D, Xu Z, Li C, Yang R, Shahidehpour M, Wu Q, et al. Economic and sustainability promises of wind energy considering the impacts of climate change and vulnerabilities to extreme conditions. Electr J 2019;32:7–12. https://doi.org/10.1016/j.tej.2019.05.013.
- [101] Ratnam EL, Baldwin KGH, Mancarella P, Howden M, Seebeck L. Electricity system resilience in a world of increased climate change and cybersecurity risk. Electr J 2020;33:106833. https://doi.org/10.1016/j.tej.2020.106833.
- [102] Thomas SR, Martínez-Alvarado O, Drew D, Bloomfield H. Drivers of extreme wind events in Mexico for windpower applications. Int J Climatol 2021;41:E2321–40. https://doi.org/10.1002/joc.6848.
- [103] van der Wiel K, Stoop LP, van Zuijlen BRH, Blackport R, van den Broek MA, Selten FM. Meteorological conditions leading to extreme low variable renewable energy production and extreme high energy shortfall. Renew Sustain Energy Rev 2019;111:261–75. https://doi.org/10.1016/j.rser.2019.04.065.

- [104] Martin N, Rice J. Power outages, climate events and renewable energy: Reviewing energy storage policy and regulatory options for Australia. Renew Sustain Energy Rev 2021;137:110617. https://doi.org/10.1016/j.rser.2020.110617.
- [105] Jordaan SM. Resilience for power systems amid a changing climate. Bull At Sci 2018;74:95– 101. https://doi.org/10.1080/00963402.2018.1436810.
- [106] Donk P, Van Uytven E, Willems P. Statistical methodology for on-site wind resource and power potential assessment under current and future climate conditions: a case study of Suriname. SN Appl Sci 2019;1:846. https://doi.org/10.1007/s42452-019-0885-6.
- [107] Huang J, Jones B, Thatcher M, Landsberg J. Temperature impacts on utility-scale solar photovoltaic and wind power generation output over Australia under RCP 8.5. J Renew Sustain Energy 2020;12:046501. https://doi.org/10.1063/5.0012711.
- [108] Ward DM. The effect of weather on grid systems and the reliability of electricity supply. Clim Change 2013;121:103–13. https://doi.org/10.1007/s10584-013-0916-z.
- [109] Cross BD, Kohfeld KE, Bailey J, Cooper AB. The Impacts of Wind Speed Trends and 30-Year Variability in Relation to Hydroelectric Reservoir Inflows on Wind Power in the Pacific Northwest. PLOS ONE 2015;10:e0135730. https://doi.org/10.1371/journal.pone.0135730.
- [110] Patt A, Pfenninger S, Lilliestam J. Vulnerability of solar energy infrastructure and output to climate change. Clim Change 2013;121:93–102. https://doi.org/10.1007/s10584-013-0887-0.
- [111] Mavromatidis G, Orehounig K, Carmeliet J. Design of distributed energy systems under uncertainty: A two-stage stochastic programming approach. Appl Energy 2018;222:932–50. https://doi.org/10.1016/j.apenergy.2018.04.019.
- [112] Balmes JR. Where There's Wildfire, There's Smoke. N Engl J Med 2018;378:881–3. https://doi.org/10.1056/NEJMp1716846.
- [113] Messier KP, Tidwell LG, Ghetu CC, Rohlman D, Scott RP, Bramer LM, et al. Indoor versus Outdoor Air Quality during Wildfires. Environ Sci Technol Lett 2019;6:696–701. https://doi.org/10.1021/acs.estlett.9b00599.
- [114] Fisk WJ. Review of some effects of climate change on indoor environmental quality and health and associated no-regrets mitigation measures. Build Environ 2015;86:70–80. https://doi.org/10.1016/j.buildenv.2014.12.024.
- [115] Xu J, Yao R, Qiu F. Mitigating Cascading Outages in Severe Weather Using Simulation-Based Optimization. IEEE Trans Power Syst 2021;36:204–13. https://doi.org/10.1109/TPWRS.2020.3008428.
- [116] Noebels M, Dobson I, Panteli M. Observed Acceleration of Cascading Outages. IEEE Trans Power Syst 2021;36:3821–4. https://doi.org/10.1109/TPWRS.2021.3071028.
- [117] Athari MH, Wang Z. Stochastic Cascading Failure Model With Uncertain Generation Using Unscented Transform. IEEE Trans Sustain Energy 2020;11:1067–77. https://doi.org/10.1109/TSTE.2019.2917842.
- [118] Nesti T, Zocca A, Zwart B. Emergent Failures and Cascades in Power Grids: A Statistical Physics Perspective. Phys Rev Lett 2018;120:258301. https://doi.org/10.1103/PhysRevLett.120.258301.
- [119] Jin Z, Wang N, Zhao J. Cascading failures of overload behaviors on interdependent networks. 2020 Asia-Pac. Int. Symp. Adv. Reliab. Maint. Model. APARM, 2020, p. 1–5. https://doi.org/10.1109/APARM49247.2020.9209434.
- [120] Thacker S, Pant R, Hall JW. System-of-systems formulation and disruption analysis for multiscale critical national infrastructures. Reliab Eng Syst Saf 2017;167:30–41. https://doi.org/10.1016/j.ress.2017.04.023.
- [121] Kong P-Y. Routing in Communication Networks With Interdependent Power Grid. IEEEACM Trans Netw 2020;28:1899–911. https://doi.org/10.1109/TNET.2020.3001759.
- [122] Ouyang M. Critical location identification and vulnerability analysis of interdependent infrastructure systems under spatially localized attacks. Reliab Eng Syst Saf 2016;154:106–16. https://doi.org/10.1016/j.ress.2016.05.007.

- [123] Bloomfield RE, Popov P, Salako K, Stankovic V, Wright D. Preliminary interdependency analysis: An approach to support critical-infrastructure risk-assessment. Reliab Eng Syst Saf 2017;167:198–217. https://doi.org/10.1016/j.ress.2017.05.030.
- [124] Tsavdaroglou M, Al-Jibouri SHS, Bles T, Halman JIM. Proposed methodology for risk analysis of interdependent critical infrastructures to extreme weather events. Int J Crit Infrastruct Prot 2018;21:57–71. https://doi.org/10.1016/j.ijcip.2018.04.002.
- [125] Guan X, Chen C. General methodology for inferring failure-spreading dynamics in networks. Proc Natl Acad Sci 2018;115:E8125–34. https://doi.org/10.1073/pnas.1722313115.
- [126] Korkali M, Veneman JG, Tivnan BF, Bagrow JP, Hines PDH. Reducing Cascading Failure Risk by Increasing Infrastructure Network Interdependence. Sci Rep 2017;7:44499. https://doi.org/10.1038/srep44499.
- [127] Heracleous C, Kolios P, Panayiotou CG, Ellinas G, Polycarpou MM. Hybrid systems modeling for critical infrastructures interdependency analysis. Reliab Eng Syst Saf 2017;165:89–101. https://doi.org/10.1016/j.ress.2017.03.028.
- [128] Shekhtman LM, Danziger MM, Havlin S. Recent advances on failure and recovery in networks of networks. Chaos Solitons Fractals 2016;90:28–36. https://doi.org/10.1016/j.chaos.2016.02.002.
- [129] Panigrahi P, Maity S. Structural vulnerability analysis in small-world power grid networks based on weighted topological model. Int Trans Electr Energy Syst 2020;30:e12401. https://doi.org/10.1002/2050-7038.12401.
- [130] Rios MA, Kirschen DS, Jayaweera D, Nedic DP, Allan RN. Value of security: modeling timedependent phenomena and weather conditions. IEEE Trans Power Syst 2002;17:543–8. https://doi.org/10.1109/TPWRS.2002.800872.
- [131] Qi J, Sun K, Mei S. An Interaction Model for Simulation and Mitigation of Cascading Failures. IEEE Trans Power Syst 2015;30:804–19. https://doi.org/10.1109/TPWRS.2014.2337284.
- [132] Fang Y-P, Pedroni N, Zio E. Comparing Network-Centric and Power Flow Models for the Optimal Allocation of Link Capacities in a Cascade-Resilient Power Transmission Network. IEEE Syst J 2017;11:1632–43. https://doi.org/10.1109/JSYST.2014.2352152.
- [133] Nakarmi U, Rahnamay-Naeini M, Khamfroush H. Critical Component Analysis in Cascading Failures for Power Grids Using Community Structures in Interaction Graphs. IEEE Trans Netw Sci Eng 2020;7:1079–93. https://doi.org/10.1109/TNSE.2019.2904008.
- [134] Noebels M, Preece R, Panteli M. AC Cascading Failure Model for Resilience Analysis in Power Networks. IEEE Syst J 2020:1–12. https://doi.org/10.1109/JSYST.2020.3037400.
- [135] Wu X, Wu D, Modiano E. Predicting Failure Cascades in Large Scale Power Systems via the Influence Model Framework. IEEE Trans Power Syst 2021;36:4778–90. https://doi.org/10.1109/TPWRS.2021.3068409.
- [136] Gupta S, Kazi F, Wagh S, Singh N. Analysis and prediction of vulnerability in smart power transmission system: A geometrical approach. Int J Electr Power Energy Syst 2018;94:77–87. https://doi.org/10.1016/j.ijepes.2017.06.033.
- [137] Hu P, Fan W-L. Mitigation strategy against cascading failures considering vulnerable transmission line in power grid. Phys Stat Mech Its Appl 2020;540:123230. https://doi.org/10.1016/j.physa.2019.123230.
- [138] Eisenberg DA, Park J, Seager TP. Linking Cascading Failure Models and Organizational Networks to Manage Large-Scale Blackouts in South Korea. J Manag Eng 2020;36:04020067. https://doi.org/10.1061/(ASCE)ME.1943-5479.0000820.
- [139] Fang Y, Pedroni N, Zio E. Optimization of Cascade-Resilient Electrical Infrastructures and its Validation by Power Flow Modeling. Risk Anal 2015;35:594–607. https://doi.org/10.1111/risa.12396.
- [140] Kamissoko D, Nastov B, Benaben F, Chapurlat V, Bony-Dandrieux A, Tixier J, et al. Continuous and multidimensional assessment of resilience based on functionality analysis for

interconnected systems. Struct Infrastruct Eng 2019;15:427–42. https://doi.org/10.1080/15732479.2018.1546327.

- [141] Yang Z, Chen Y, Marti J. Modelling cascading failure of a CPS for topological resilience enhancement. IET Smart Grid 2020;3:207–15. https://doi.org/10.1049/iet-stg.2019.0169.
- [142] Arrighi C, Pregnolato M, Castelli F. Indirect flood impacts and cascade risk across interdependent linear infrastructures. Nat Hazards Earth Syst Sci 2021;21:1955–69. https://doi.org/10.5194/nhess-21-1955-2021.
- [143] Mazur C, Hoegerle Y, Brucoli M, van Dam K, Guo M, Markides CN, et al. A holistic resilience framework development for rural power systems in emerging economies. Appl Energy 2019;235:219–32. https://doi.org/10.1016/j.apenergy.2018.10.129.
- [144] Arghandeh R, von Meier A, Mehrmanesh L, Mili L. On the definition of cyber-physical resilience in power systems. Renew Sustain Energy Rev 2016;58:1060–9. https://doi.org/10.1016/j.rser.2015.12.193.
- [145] Sukhwani V, Shaw R, Mitra BK, Yan W. Optimizing Food-Energy-Water (FEW) nexus to foster collective resilience in urban-rural systems. Prog Disaster Sci 2019;1:100005. https://doi.org/10.1016/j.pdisas.2019.100005.
- [146] Sharifi A, Yamagata Y. Principles and criteria for assessing urban energy resilience: A literature review. Renew Sustain Energy Rev 2016;60:1654–77. https://doi.org/10.1016/j.rser.2016.03.028.
- [147] Mendizabal M, Heidrich O, Feliu E, García-Blanco G, Mendizabal A. Stimulating urban transition and transformation to achieve sustainable and resilient cities. Renew Sustain Energy Rev 2018;94:410–8. https://doi.org/10.1016/j.rser.2018.06.003.
- [148] Cantelmi R, Di Gravio G, Patriarca R. Reviewing qualitative research approaches in the context of critical infrastructure resilience. Environ Syst Decis 2021;41:341–76. https://doi.org/10.1007/s10669-020-09795-8.
- [149] Meerow S, Newell JP, Stults M. Defining urban resilience: A review. Landsc Urban Plan 2016;147:38–49. https://doi.org/10.1016/j.landurbplan.2015.11.011.
- [150] Ribeiro PJG, Pena Jardim Gonçalves LA. Urban resilience: A conceptual framework. Sustain Cities Soc 2019;50:101625. https://doi.org/10.1016/j.scs.2019.101625.
- [151] Nelson K, Gillespie-Marthaler L, Baroud H, Abkowitz M, Kosson D. An integrated and dynamic framework for assessing sustainable resilience in complex adaptive systems. Sustain Resilient Infrastruct 2020;5:311–29. https://doi.org/10.1080/23789689.2019.1578165.
- [152] Cariolet J-M, Vuillet M, Diab Y. Mapping urban resilience to disasters A review. Sustain Cities Soc 2019;51:101746. https://doi.org/10.1016/j.scs.2019.101746.
- [153] Rus K, Kilar V, Koren D. Resilience assessment of complex urban systems to natural disasters: A new literature review. Int J Disaster Risk Reduct 2018;31:311–30. https://doi.org/10.1016/j.ijdrr.2018.05.015.
- [154] Salimi M, Al-Ghamdi SG. Climate change impacts on critical urban infrastructure and urban resiliency strategies for the Middle East. Sustain Cities Soc 2020;54:101948. https://doi.org/10.1016/j.scs.2019.101948.